Author's response to reviews

Title: Can Magnetic Resonance Imaging Findings Predict the Degree of Knee Joint Laxity in Patients Undergoing Anterior Cruciate Ligament Reconstruction?

Authors:

Moon Jong Chang (moonjongchang@gmail.com)
Chong Bum Chang (ccbknee@gmail.com)
Ja-Young Choi (drchoi01@gmail.com)
Min Soo Je (peteje@naver.com)
Tae Kyun Kim (osktk@snubh.org)

Version: 4 Date: 9 June 2014

Author's response to reviews:

MS: 4699595101166082
Research article
Can Magnetic Resonance Imaging Findings Predict the Degree of Knee Joint Laxity in Patients Undergoing Anterior Cruciate Ligament Reconstruction?
Moon Jong Chang, Chong Bum Chang, Ja-Young Choi, Min Soo Je and Tae Kyun Kim
BMC Musculoskeletal Disorders (Section: Clinical diagnostics and imaging)

We notice that this study is a retrospective study performed on prospectively collected data. We need a final confirmation as to whether appropriate consent/permission was obtained for access to this prospectively collected data. Please ensure a statement is included in the manuscript to confirm that you had consent to access and use this data.

-> We added the description in the paragraph 1 of the Method.